Author’s response to reviews

Title: Dissemination and implementation of suicide prevention training in one Scottish region

Authors:

Linda Gask (Linda.Gask@manchester.ac.uk)
Gillian D Lever-Green (g.d.lever-green@manchester.ac.uk)
Rebecca Hays (Rebecca.Hays@manchester.ac.uk)

Version: 2 Date: 22 July 2008

Author’s response to reviews: see over
Dear Editor

Education and training initiatives are frequently espoused as necessary and timely in health services, but are rarely evaluated and the factors governing whether they succeed or fail may remain unclear. We hope that you will consider our paper, which is concerned with the process of dissemination of one such training intervention in one Scottish region, for publication in your journal. We have attempted to elucidate the processes involved in dissemination and implementation of training in suicide risk assessment and management, which has been national policy in both England and Scotland for some years.

Yours truly

Linda Gask